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PACIFIC REGION CCG VESSEL -POST CRUISE REPORT

NAME OF SHIP/PLATFORM:   JP TULLY
DATE:

FROM:
  Sept 22 2003

TO:  Sept 30 2003
SCIENCE CRUISE NUMBER:
IOS 2003-31

SHIP’S PATROL NUMBER:

CHIEF SCIENTIST[S]:  David Mackas
ADDITIONAL SCIENTIFIC PERSONNEL:



	
MALE [NAMES]
	FEMALE[NAMES]

	Doug Yelland (IOS)
	Moira Galbraith (IOS)

	Doug Moore (IOS)
	Melissa Hennekes (IOS)

	Darren Tuele (IOS)
	Tanja Gschwind (IOS visitor)

	Doug Anderson (IOS)
	Stephanie King (IOS)

	Wei Yan (Nanjing Univ, IOS visitor)
	

	Kelly Field (volunteer)
	


AREAS OF OPERATION West coast of Vancouver Island, Juan de Fuca Strait to southern Queen Charlotte Sound, Queen Charlotte Strait
CRUISE OBJECTIVE/OBJECTIVES:
-Biological oceanography time series (CTD, nutrients, zooplankton), west coast Vancouver Island 
- Deploy `whale-listening`hydrophone mooring, continental slope at N end of Vanc Island

- Nestucca Project sampling of oceanographic conditions affecting the Triangle Island seabird colony 

DAYS ALLOCATED:
9


DAYS OF OPERATION: 
9

TIME LOST DUE TO WEATHER: Excellent weather throughout the cruise. A few minor delays & slowdowns of gear deployment, totalling less than 4 hrs.

TIME LOST DUE TO OTHER CAUSES: Numerous minor delays totalling about 30 hours. See `Problems`sections below.
RESULTS:

We completed plankton and water property measurements along 4 outer coast survey lines (extending seaward from Barkley Sound, Estevan Point, Brooks Peninsula, and the Scott Islands), plus 4 inner coast sampling stations in Queen Charlotte/Johnstone Straits, and `whale` hydrophone mooring deployed near S. Dellwood ODAS. 
Zooplankton samples showed a distinct gradient in community composition between southern vs northern Vancouver Island. Southern sites were dominated by cold water continental shelf taxa. Northern Vancouver Island was dominated by warmer water oceanic taxa.

We encountered an extremely intense dinoflagellate (red tide) bloom over the continental slope off Estevan Point. About 20 km cross-shore width, probably  >50 km alongshore.  Surface water was distinctly red-brown in color. Species were a mix of several dinoflagellate genera. Not sure yet if they were toxic species, but we have preserved sample for taxonomic analysis.
PROBLEMS [SCIENTIFIC GEAR AND OPERATIONS]:

CTD/rosette: A series of glitches with data transmission from and to the CTD rosette. Most troublesome was unreliable bottle trip. Eventually solved by swapping SBE deck unit, but many hours spent testing other alternatives. Part of the difficulty in diagnosis was multiple causes: defective rosette pylon on the previous cruise, poor data transfer through winch cable on deep casts. Shortening (500m) and re-terminating the cable gave partial improvement of the deck unit performance 
BIONESS: Also had wet/worn data cable problems during later tows with the BIONESS. This problem was temporarily bypassed by switching net release mode to manual, but need more shore-side work.

Bongos: Occasional tangling and twisting of towing-point hardware. Swivel & shackle combination needs redesign.
Oxygen kit: Bad (although repeatable) standardization of thiosulfate solution lead to all oxy concentration estimates being low by factor of about two. Apparently this was also a problem on the previous Line P cruise – not clear why it was not resolved prior to this cruise. URGENT need for proper standardization, and recalculation of all data.

 SUCCESSES [SCIENTIFIC]:
See results. High grades in general to the scientific party for good work and good on-board attitude. We completed about 90% of our intended sampling program, and obtained a good description of biological, chemical, and physical oceanographic conditions off the west coast in late summer/early autumn of 2003. Nutrient auto-analyzer worked well throughout the cruise. Congrats to new sea-going AA tech Melissa Hennekes, and IOS volunteer and new PBS employee Kelly Field. 
PROBLEMS [SHIP’S EQUIPMENT/

OPERATIONS/PLATFORM SUITABILITY]:

Ships equipment problems: Various problems with hydraulic connections, hydraulic supply, and spooling gear on the scientific winches led to occasional minor delays and lower than desired payout and retrieval of wire. Also, the small electric winch (`Tully Tugger`) used to lift rosette stabilizer weight was burned out in previous cruise; replacement was on board but not found until mid-way through cruise

Training & seasoning of staff: Many (in my opinion too many) new hands in both ships and scientific crew. On-the-job training and familiarization with operating procedures slowed down operations, especially during the first 2-3 days. After that, the deployment and sampling routines went well (combination of quick & motivated learners, watchful teachers and good weather), but we might have had serious difficulty completing the program safely if the weather had been marginal.

Simrad 12kHz echosounder: The lab echosounder had intermittently very weak bottom returns from depths >500m, especially in choppy sea state. This also was a problem on the previous (Line P) survey. I think that much of the problem is due to bio-fouling of the hull-mount transducer. Apparently these were last inspected & cleaned about 18 months ago. A dense coating of algae, barnacles, mussels etc. would trap air bubbles and block sound transmission between the water and the transducer head. The `monochrome display`, seen as a symptom on the previous cruise, appears to be normal: only the strongest `heard` sound pulse is color coded
SUCCESSES [SHIP]:

As always, the TULLY is an excellent platform for our work. Lab and deck layout were excellent. All of the crew (bridge, deck, engineering, and galley) were friendly and helpful, and the new hands were performing above expectation by the second half of the cruise. 

SAFETY CONCERNS:
None, other than the late appearance of the Tully Tugger winch

HAZARDOUS OCCURRENCES:

CTD/rosette laid on bottom at station LBP6. Cause was steep bottom slope, plus unreliable operation of 12 kHz echosounder, plus lack of communication at watch change. Damage to approx 100m of CTD cable (removed and again re-terminated).
EVENT LOG:

Monday 22 Sept: 

0900 Mobile crane offloads spare CTD winch, puts on BIONESS winch. 

1000-1700: Load & deck test scientific gear. 
-Replacement bongo winch installed in AM, various problems with hydraulic line connections, fixed late PM. Calculated bongo winch counter calibration  from sheave & wire diameters (formula and instructions wrong; recalibrated later based on measure wire out).
-Tully Tugger electric winch off for repairs, crew initially unable to find replacement unit. Agree to use capstan and rope as substitute
1630-1700: safety briefing of science party

1830-2000: Leave IOS wharf. Test and demo of rosette operation, Saanich Inlet station 

StnID
Lat

Long

Depth
Work

SI01

48 37.1
123 30.0  
217

CTD-rosette

2000: Depart for west coast Vancouver Island.

Tuesday 23 Sept:

Sampling stations (water properties, zooplankton bongo & BIONESS tows for IOS and UVic) off southern Vancouver Island

Time(UTC)StnID
Lat

Long

z(m)
Work

1450

C2(LBA2)
48 25.4
125 07.8
115
CTD, 0-110 bongo

1710

C1

48 28.93
125 15.3
150
CTD/R, 0-145 & 0-50 (UVic) bongo

1856

C3(LBA3)
48 23.45
125 20.8
120
CTD, 0-115 bongo

2145

B8(LCB3)
48 34.5
125 30.0
120
CTD/R, 0-115 & 0-50 (UVic) bongo

2333

B7(LCB4)
48 32.0
125 35.5
76
CTD, 0-70m bongo

0241

A2

48 22.7
126 03.8
400
CTD, 0-250m bongo

0525

LC11

48 18.9
126 26.8
1470
CTD/R, 0-50 (UVic) bongo
Wednesday 24 Sept.

Continue sampling off southern Vancouver Island

0810

A4(LC12)
48 15.0
126 40.0
2524 BION (night),CTD/R,0-250bongo

1524

LC7

48 32.96
126 00.5
125
CTD/R

1638

A1(LC8)
48 29.45
126 07.1
197
CTD, bongo

1835

A2(LC9)
48 25.94
126 13.7
660
CTD/R, bongo

2154

A4(LC12)
48 15.0
126 40.0
2500
BION(day) 

0017

LC6

48 36.46
125 54.0
95
CTD, bongo

0348

LC5

48 39.94
125 47.4
65
CTD/R, bongo

0537

LC3

48 46.95
125 34.24
130
CTD

0642

LC1

48 50.44
125 27.73
90
CTD/R

Thursday 25 Sept.

Complete work in southern Vancouver Island (La Perouse) area, steam north to Estevan Point line

0852

D1(LD2) 
48 58.35
125 47.05
42
CTD, 2 bongos (1 for UVic)

1036

D2(LD4) 
48 53.16
125 57.0
60 
CTD, 2 bongos (1 for UVic)

Enroute to LG line, 500m of CTD wire removed, re-terminated

1722

LG1

49 20.5
126 35.0
55
CTD/R

1910

LG2

49 18.7
126 38.1
105
CTD/R, bongo

2015

LG3

49 15.0
126 43.7
112
CTD

2116

LG4

49 11.3
126 49.4
146
CTD/R, bongo

2241

LG5

49 07.4
126 55.3
258
CTD

Very dense red tide observed near LG6

0005

LG6

49 03.5
127 01.2
866
CTD

0135

LG7

48 59.4
127 07.2
1061
CTD/R, bongo

0503

LG8

48 55.30
127 13.30
2070
CTD

0655

LG9

48 51.2
127 19.4
2059
CTD/R, bongo

Problems with rosette bottle trips re-appear at outer LG stations

Friday 26 Sept.

Steam north to Brooks Peninsula (LBP) station line

1724

LBP8

49 48.6
128 16.8
2050 m
CTD

1940

LBP7

49 52.4
128 11.2
2000 m
CTD/R, 2 bongo

2245

LBP6

49 56.2
128 05.5
700m

CTD

(hit bottom with CTD cast, damaged & removed approx 100m of cable)

0151

LBP5

50 00.0
128 00.0
1230m
CTD/R, bongo

0434

LBP4

50 01.4
127 58.1
1106 m
CTD

0555

LBP3

50 03.2
127 55.3
172 m
CTD/R,2 bongo

0726-27
LBP2

50 04.0
127 54.2
105m

CTD, bongo

0825-27
LBP1

50 04.8
127 52.8
32m

CTD
Saturday 27 September

Steam north to Cape Scott mooring & BIONESS day-night pair of stations

2000 

Deploy whale hydrophone mooring approx 5 miles NW of ODAS met buoy (50 57.496  129 59.609)
2200-0030 `ODAS
50 53

129 58
2000+
day BION (2 tries)

0210

ODAS
50 53.11
130 00.4
2140

CTD/R, bongo

0445

ODAS






night BION

Sunday 28 Sept

Triangle Island station line

0814

CS1

50 34.9
129 41.5
2107

CTD, 2 bongos 

1132

CS2

50 41.3
129 28.0
1900

CTD/R, bongo 

1454

CS3

50 45.6
129 20.0
200

CTD, bongo 

1626

CS4

50 49.2
129 13.0
100

CTD/R, bongo 

(About 4 hours of bad data from loop salinometer – something stuck in cell)

1818

CS5

50 56.0
129 00.0
64

CTD, bongo 


1943

CS6

51 00.0
128 52.1
65

CTD/R, bongo

2111

CS7

51 04.5
128 44.0
65

CTD, bongo

2231

CS8

51 08.5
128 36.0
138

CTD/R, bongo

0008

CS9

51 12.5
128 28.0
195

CTD, 2 bongos 

0159

CS10

51 16.5
128 20.0
78

CTD/R, bongo

Steam to entrance of Queen Charlotte Strait

0535

CPE1
51 00.0
127 50.0
150

CTD/R, 2 bongos

Monday 29 Sept.

Stations for Dario Stucchi in Queen Charlotte Strait

0902

QCS5
50 49.43
127 21.61
158

CTD/R, bongo 

1044

QCS3
50 43.76
127 15.0
210

CTD

1255

QCS1
50 40.06
126 46.29
150

CTD/R

1426

JS1

50 33.0
126 46.0
460

CTD

Steam to IOS via Johnstone Strait

Tuesday 30 September

Arrive IOS 0730 to offload.
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